%
% Glossar:
%

\chapter{Glossar}

\begin{description}

    \item [Frontend] \hfill \\
        \quotecite{Front-end website is a combination of
hypertext markup language (HTML), cascading style sheets (CSS) and JavaScript that generates pages that can be
viewed and interacted directly with the user.} \cite[S. 294]{bib:udjaja}.

    \item [TYPO3-Backend] \hfill \\
        \quotecite{The backend's main role is to enable users to create and publish
            content for their site.
            The backend is also used to configure a TYPO3 installation. Domains,
            languages and other information that determine how a site behaves are managed
            via the backend. Tasks such as adding backend users and managing
            third-party extensions also take place in the backend.
        } \cite{bib:typo3-docs-getting-started}.

    \item [TYPO3-Extension] \hfill \\
        \quotecite{Developed by the community, extensions provide a range of solutions that
          help extend TYPO3} \cite{bib:typo3-docs-getting-started}.

    \item [TYPO3-Plugin] \hfill \\
        \quotecite{Plugins are a specific type of content elements [...]
          Plugins often use additional database tables which contain records which
          are dynamically displayed via the plugin - often in a list view, a
          single view, optionally with pagination and search functionality.
          An extension may provide several plugins, each with a dedicated function,
          such as the list view.} \cite{bib:typo3-docs-content-elements-plugins-introduction}

    \item [Weinland-Mosel-Mitglied] \hfill \\
      Ein Weinland-Mosel-Mitglied ist ein Mitglied des Moselwein e.V.
      Diese Mitglieder sind i.d.R. Weingüter und weinproduzierende Unternehmen.
      Mitglieder profitieren von einer Reihe an
      Vorteilen, wie beispielsweise einer Auflistung auf Weinland Mosels Internetauftritt.
      Durch diesen Auftritt sind bereits Stammdatensätze vorhanden.
      Nur durch eine Teilnahme an einer Jahresauswahlprobe ist ein Unternehmen kein Mitglied.

    \item [Flight] \hfill \\
      Pro Tisch einer Jahresauswahlprobe werden sechs Weine gleichzeitig eingeschenkt und bewertet.
      Das ist ein Flight. Pro Tisch gibt es mehrere aufeinanderfolgende Flights.

\end{description}
